Author’s response to reviews

Title: Barriers to the Acceptance of Electronic Medical Records by Physicians From Systematic Review to Taxonomy and Interventions

Authors:

Albert Boonstra (albert.boonstra@rug.nl)
Manda Broekhuis (h.broekhuis@rug.nl)

Version: 3 Date: 5 August 2010

Author’s response to reviews:

Dear members of the editorial production team of BMC health services research

We have received your mail of 4 August 2010 in good order and we have addressed the major revisions.

1) Both authors read and approved the final manuscript, which is indicated in the text.
2) References are now in numerical order.
3) Table 1 is reduced to 3 pages.

Minor revisions:
1) The table box around the title is removed.
2) The title is now in sentence case.
3) Highlighting is now removed from the manuscript.
4) Placement text of figures and tables is NOT yet removed, because we were not sure how the production team can identify the right place for figures or tables.
5) Competing interests section is now before author’s contributions.
6) Reference details of [28] are now included.
7) Visible vertical lines are now removed from the tables.
8) Image files of figures does not include titles.
9) We have checked the whole document.